COS 341, September 21, 2001 Handout Number 2

Homework Grading Policy

Homework grading.

• Each homework problem will be worth the same amount, unless otherwise noted. • All homework assignments will be equally weighted.

• We will try to return homework and post solutions one week after the due date.

Late homework.

• Homework is due Wednesdays during class.

• Late homework submitted before 5 PM on the Friday following its due date will be penalized 50%.

• Homework will not be accepted after Friday 5 PM except in extreme circumstances. • Late homework can be placed in Carl Kingsford’s mailbox on the 2nd floor of the CS building.

Office hours.

• Jessica Fong: Tuesdays, 10:30–noon, room 413. • Carl Kingsford: Mondays, 1:30–3:00, room 223.